Author’s response to reviews

Title: Hidden intra-mandibular carcinoma cuniculatum appearing in a patient with metastatic prostate cancer : A case Report

Authors:

Elyette Broly (elyette.broly@gmail.com)

Philippe Barthélémy (philippe.barthelemy@chru-strasbourg.fr)

Saït Ciftci (sciftci@strasbourg.unicancer.fr)

Christian Borel (cborel@strasbourg.unicancer.fr)

Martin Broly (martin.broly@gmail.com)

Catherine-Isabelle Gros (catherine-isabelle.gros@chru-strasbourg.fr)

Luc Marcellin (luc.marcellin@chru-strasbourg.fr)

Fabien Bornert (fabien.bornert@chru-strasbourg.fr)

Version: 2 Date: 14 Mar 2019

Author’s response to reviews:

Minor Revisions - Response letter

Sadeq Ali Al-Maweri (BMC Oral Health’s editor) :

Thank you for your technical comments.

All requested changes have been done. Please find them in the clean version of my manuscript

Title

The article "an" is not appropriate so please delete it from the title.

Title has been modified.

Abstract:
- In the background of the abstract, the aim of the report was not properly formulated, so please try to paraphrase it. You may write: We report a rare case of ............

The aim of the report has been properly formulated: « We report a rare case of carcinoma cuniculatum in a context of metastatic prostate cancer. No case was previously described. » (Abstract section, background, page 5, line 17).

- The conclusion needs to be modified to capture the presented case. You may write this beautifully written sentence instead: "This case recalls that any persistent tooth pain affecting cancer patients requires a thorough review to exclude any secondary primary cancers or any metastasis of the oral cavity and more specifically in jawbones."

Conclusion has been modified. The sentence proposed: "This case recalls that any persistent tooth pain affecting cancer patients requires a thorough review to exclude any secondary primary cancers or any metastasis of the oral cavity and more specifically in jawbones." has been added instead (Abstract section, Conclusion, Page 5, line 59).

**Background**

In page 7, lines 39-40 the sentence "For this patient, the mandibular lesion was diagnosed one year after treatment of metastatic prostate cancer." does not fit well here and so I recommend to get it deleted.

All requested changes have been added. The sentence: "For this patient, the mandibular lesion was diagnosed one year after treatment of metastatic prostate cancer." has been deleted. (Manuscript section, Background, Page 7, lines 39-40).

Also please do double-check typo errors throughout the text and ensure to follow the journal's guidelines.

The manuscript has been double-checked and revised according to journal’s guidelines.